Author's response to reviews

Title: Therapeutic effects of STAT3 decoy oligodeoxynucleotide on human lung cancer in xenograft mice

Authors:

Xulong Zhang (zhxlwl@yahoo.com.cn)
Jian Zhang (zhangj65@sdu.edu.cn)
Lihua Wang (lhwang@ncifcrf.gov)
Haiming Wei (ustcwhm@ustc.edu.cn)
Zhigang Tian (tzg@ustc.edu.cn)

Version: 6 Date: 1 August 2007

Author's response to reviews:

Dear editor,
Thank you very much for your response to our manuscript (MS number 3170052101292317). As your requirement, we revised our MS and figures.
Sincerely,
Jian Zhang, & Zhiguang Tian, M.D., Ph.D.

Institute of Immunopharmacology and Immunotherapy
Shandong University,
Jinan 250012, China.

E-mail: zhangj65@sdu.edu.cn; tzg@ustc.edu.cn;
Tel: 86-531-88381980
Fax: 86-531-88383782